Author's response to reviews

Title: A randomised clinical trial on comprehensive geriatric assessment and intensive home follow-up after hospital discharge: the transitional care bridge

Authors:

Bianca M Buurman (b.m.vanes@amc.uva.nl)
Juliette L Parlevliet (j.l.parlevliet@amc.uva.nl)
Bob AJ van Deelen (bvdeelen@amc.uva.nl)
Rob J de Haan (r.j.dehaan@amc.nl)
Sophia E de Rooij (s.e.derooij@amc.uva.nl)

Version: 2 Date: 10 October 2010

Author's response to reviews: see over
Amsterdam, October 9, 2010

Dear Dr Marshall,

Thank you for reviewing our study protocol entitled ‘A randomised clinical trial on comprehensive geriatric assessment and intensive home follow up after hospital discharge: the Transitional Care Bridge’.

We had to make a few adaptations to the study protocol, which are hereby addressed. We hope this information is sufficient for further publication of the study protocol.

Comments

1. Please provide the trial registration number at the foot of the Abstract
   We have added the trial registration number of the trial at the foot of the abstract. The trial is registered in the Netherlands Trial Register (NTR).

2. Please be sure to provide details about the funding within the Acknowledgements section of manuscript.
   In the acknowledgement section we added information on the grant and the grant number.

3. Please correct the formatting of table 2.
   We have corrected the formatting of table 2.

With kind regards, also on behalf of the other authors
Bianca Buurman, MSc